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Status 
 The intensity for RHIC fill is increased modestly, to 1.7*1011.  

 No RF station B since Friday night. This caused beam loss on the ramp. The 

Booster scraping is relaxed to get enough intensity for RHIC fill. Slightly 

larger vertical emittance is observed for some fills.  

 A limit switch for AGS polarimeter target control failed on Thursday. No 

polarization measurement until Friday night. It has been fixed by access on 

Friday.   

 No Au development this week, so no need of Au setup in the AGS today. 

 NSRL and AMS runs are starting in 11 days (Mar. 14). Preparation is 

underway.  


